**Author’s response to reviews**

**Title:** The effectiveness of a Value-based EMotion-cognition-Focused educational programme to reduce diabetes-related distress in Malay adults with Type 2 diabetes (VEMOFIT): study protocol for a cluster randomised controlled trial

**Authors:**

Boon How Chew (chewboonhow@gmail.com)

Rimke Vos (R.C.Vos-3@umcutrecht.nl)

Sazlina Shariff Ghazali (drszazsg@gmail.com)

Nurainul Hana Shamsuddin (nurainul@upm.edu.my)

Aaron Fernandez (aaronhkl@gmail.com)

Firdaus Mukhtar (drfircbt@gmail.com)

Mastura Ismail (drmastura@moh.gov.my)

Azainorsuzila Mohd Ahad (suzilahad@hotmail.com)

Narayanan Sundram (narayanansundram@gmail.com)

Siti Zubaidah Mohd Ali (dwnsiti@yahoo.com.my)

Guy Rutten (G.E.H.M.Rutten@umcutrecht.nl)

**Version:** 2 Date: 03 Mar 2017

**Author’s response to reviews:**

Response to the Editor Comments

Editor commented:

Please revise your manuscript to include all the following sections under the heading 'Declarations':

- Ethics approval and consent to participate
- Consent for publication
Availability of data and material

Competing interests

Funding

Authors' contributions

Acknowledgements

Authors' information (optional)

Response:

Thank you for your comments and the opportunity to improve our manuscript. The missing sections, namely the Consent for publication and Authors' information, have been added in the revised manuscript.

At page 23: Consent for publication- Not applicable.

At page 24: Authors' information- BHC, SSG and NHS are academic family medicine specialists working in Universiti Putra Malaysia (UPM). AF and FM are psychiatrist and psychologist, respectively, in UPM. MI, AMA, NNS and SZMA are family medicine specialists at the participating health clinics in Negeri Sembilan. RCV is an assistant professor, and a clinical epidemiologist at Julius Center for Health Sciences and Primary Care, University Medical Center Utrecht. GEHMR is a professor in primary care diabetology at Julius Center for Health Sciences and Primary Care, University Medical Center Utrecht, Department of General Practice, and a general practitioner in The Netherlands.